Author’s response to reviews

Title: Childhood MMR Vaccination and the Incidence Rate of Measles Infection: A Ten Year Longitudinal Cohort Study of American Children born in the 1990s

Authors:

David Geier (davidallengeier@comcast.net)
Janet Kern (jkern@dfwair.net)
Mark Geier (mgeier@comcast.net)

Version: 3 Date: 01 Sep 2019

Author’s response to reviews:

Dear Editor,

Thank you for the opportunity to revise our manuscript "Childhood MMR Vaccination and the Incidence Rate of Measles Infection: A Ten Year Longitudinal Cohort Study of American Children born in the 1990s". The suggested revisions were addressed, and changes were made to the manuscript accordingly. A point-by-point response is included. We submitted a single, clean version.

Comment 1: There is textual overlap mainly exists in the Discussion section. If there is overlap in the Methods section, please ensure to summarize the methods and cite the source.

Response to comment: We re-phrased the Discussion section to remove any overlap. We did not find overlap in Methods section.

Comment 2: We note that you have included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Response to comment: We put NA.
Comment 3: In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response to comment: We revised the MS accordingly.

Comment 4: Please represent authors’ names using their full initials, not their full name, in the Authors’ Contributions section.

Response to comment: We revised the MS accordingly.

Comment 4: Please remove the STROBE checklist from your submission’s supplementary files.

Response to comment: We revised the submission accordingly.